Author’s response to reviews

Title: Association between second-generation antipsychotics and newly diagnosed treated diabetes mellitus: does the effect differ by dose?

Authors:

Marianne Ulcickas Yood (muyood@muyood.com)
Gerald N DeLorenze (gerald.n.delorenze@kp.org)
Charles P Quesenberry Jr (Charles.Quesenberry@kp.org)
Susan A Oliveria (susano@episource.org)
Ai-Lin Tsai (ailin.tsai@kp.org)
Edward Kim (edward.kim@novartis.com)
Mark J Cziraky (MCziraky@healthcare.com)
Robert D McQuade (Robert.McQuade@otsuka.com)
John W Newcomer (JNewcomer@med.miami.edu)
Gilbert J L'Italien (litalieg@bms.com)

Version: 6 Date: 3 December 2011

Author’s response to reviews:

Dear BMC Psychiatry Editorial Team:

Attached is the revised version of the manuscript which now includes IRB information.

Please let me know if there are additional details or questions that come up.

Sincerely,

Marianne Ulcickas Yood